Author's response to reviews

Title: Epidemiology of Human Immunodeficiency Virus-1 and Hepatitis B co-infections and risk factors for acquiring these infections in the Fako Health District

Authors:

Lauren Shevell (lauren.shevell@gmail.com)
Henry D Meriki (merilonga@yahoo.com)
Fidelis Cho-Ngwa (chongwa_ub@yahoo.co.uk)
Crystal Fuller (cf317@columbia.edu)

Version: 2  Date: 6 April 2015

Author's response to reviews:

- Line numbering and page numbers have been included in the manuscript
- The specific name of the ethics committee "Cameroon National Ethics Committee has been included on line 144. The subheading "consent" has been changed to "Ethical approval and consent"
- The postal address (P.O. Box 63) of the submitting author has been included on line 18
- The abstract has also been formatted as requires and methods added.